Author's response to reviews

Title: Bone mineral density of the proximal femur after hip resurfacing arthroplasty: 1-year follow-up study.

Authors:

Arja Hakkinen (arja.hakkinen@jyu.fi)
Håkan Borg (hakan.borg@mmtnordic.se)
Mikko Hakulinen (Mikko.Hakulinen@kuh.fi)
Jukka Jurvelin (jukka.jurvelin@uef.fi)
Esa Anttila (esa.anttila@ksshp.fi)
Tapani Parviainen (tapani.parviainen@ksshp.fi)
Ilkka Kiviranta (ilkka.kiviranta@helsinki.fi)

Version: 5 Date: 28 April 2011

Author's response to reviews:

Major revisions (we require the author to make these changes)

COMMENT 1: Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Answer: Revised

COMMENT 2: Abstract: This should be identical in the manuscript and the submission system.

Answer: this has been checked/ copied from the formatted version

COMMENT 3: Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Answer: Thank you

COMMENT 4: Tables: please remove the visible vertical lines from your tables

Answer: Revised as suggested

COMMENT 5: Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number. It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:
http://www.biomedcentral.com/info/ifora/figures
Answer: Figure number deleted.

COMMENT 5: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Answer: Some minor corrections are done. e.g. anteriorposterior changed to anteroposterior